ORIGINAL RESEARCH Adverse Drug Reactions Related with Antibiotic Medicines in Malawi: A Retrospective Analysis of Prevalence and Associated Factors

Francis Kachidza Chiumia^[b], Frider Chimimba¹, Happy Magwaza Nyirongo¹, Elizabeth Lusungu Kampira², Adamson Sinjani Muula³, Felix Khuluza^[D]

Department of Pharmacy, School of Life Sciences and Allied Health Professions, Kamuzu University of Health Sciences, Blantyre, Malawi; ²Department of Medical Laboratory Sciences, School of Life Sciences and Allied Health Professions, Kamuzu University of Health Sciences, Blantyre, Malawi; ³Department of Community and Environmental Health, School of Global and Public Health, Kamuzu University of Health Sciences, Blantyre, Malawi

Correspondence: Felix Khuluza, Email fkhuluza@kuhes.ac.mw

Objective: We aimed to assess the occurrence and characteristics of antibiotic-associated adverse drug reactions (ADRs) in Malawi.

Methods: We retrospectively reviewed 304 patient records from medical wards in three hospitals in Southern Malawi. A global trigger tool was applied for the detection of suspected ADRs, and we used the Naranjo scale, the World Health Organization classification and the Schumock and Thornton scale for causality, seriousness and preventability assessment respectively. ADRs were also further characterized according to anatomical systems. Statistical analysis was done in STATA 14.1. The Chi-square test was used to determine the association between categorical variables and logistic regression analysis was used to measure the strength of the association between various independent variables and the occurrence of ADRs.

Results: Suspected ADRs were detected in 24% (73/304) of patients, of which 1.4% were definite, 15.1% were probable and 83.6% were possible ADRs. Most of the sADRs were gastrointestinal events (42.5%), followed by: musculoskeletal (26.3%); cardiovascular (16.3%); central nervous system (13.8%; and urinary events (1.3%). About 27% of the sADRs were serious events such as convulsions. The geriatric age group (\geq 65 years) was more likely to experience sADRs as compared to the younger age group, with an adjusted odds ratio (aOR) of 4.53, 95% CI (2.21-9.28), P<0.001. Patients taking more than one antibiotic medicine had a higher risk of developing sADRs as compared to patients who were administered one type of antibiotic medicine, aOR 2.14, 95% CI (1.18-3.90), p < 0.012. A long hospital stay of >3 days was associated with a higher risk of sADRs with aOR of 5.11, 95% CI (2.47-10.55), p < 0.001 than those who stayed ≤ 3 days in the hospital.

Conclusion: We found a higher prevalence of serious sADRs associated with antibiotic medicines than reported elsewhere. This may, among others, contribute to high patient mortality, poor treatment adherence, antibiotic resistance and increased cost of care.

Plain Language Summary:

What is already known and why we did the study?

- Most health care workers and patients are less likely to voluntarily report suspected adverse drug reactions in low- and middleincome countries such as Malawi.
- Studies have revealed a high usage of antibiotic medicines in Malawi, but there is limited data on the associated adverse drug reactions.

What did we do?

- We assessed the occurrence and characteristics of ADRs associated with antibiotics. What are the new findings?
- We found a higher prevalence (24%) of adverse drug reactions associated with antibiotic therapy than reported elsewhere using the global trigger tool.
- About 27.4% of the events were serious ADRs such as convulsions, arrhythmia and hypotension.
- We observed a higher rate of convulsions which could be a potential safety signal.

What do the new findings imply?

- The high prevalence of serious ADRs leads to complicated treatment strategies and contribute to patient mortality, poor treatment adherence and antibiotic resistance.
- ADR risk factors need to be considered when prescribing and monitoring patients on antibiotic therapy.

Keywords: adverse drug reactions, antibiotic medicines, pharmacovigilance, global trigger tool, Malawi

Background

Adverse drug reactions(ADRs) have a serious negative impact on patients and the healthcare system.¹ Globally, about 24% of elderly patients² and 8% of patients receiving primary health care in Europe and the United States of America experience at least one ADR.³ About 5–8% % of ADRs are serious⁴ and at least 1% of ADRs lead to patient death.⁵ ADRs have also been reported to cause hospitalization or lead to prolonged hospitalization in almost 10% of patients.^{6,7} Usually, ADRs are managed through switching of therapy, in some cases prescribing new medicines in order to counter the adverse effects, thus contributing to increased total cost of treatment.⁸ The annual total cost of ADR treatment is estimated at \$30 billion in the USA⁹ and £2.2 billion in the UK.¹⁰

Antibiotics are among the most prescribed medicines in low- and middle-income countries (LMICs) with almost 50% of prescriptions containing at least one antibiotic medicine.^{11,12} By 2030, the antibiotic consumption rate is estimated to increase by 200% if no policy change is effected in LMICs.¹³ This is a major concern because as much as antibiotic medicines benefit the public especially in settings where the burden of infectious diseases is high, overuse of antibiotics potentially increases the incidence of associated risks such as ADRs and antibiotic resistance.^{14,15} The majority of antibiotic-related ADRs are clinically significant and require additional medical attention.¹⁶ Fatal or life-threatening ADRs such as jaundice, thrombocytopenia and difficulty in breathing have also been reported to be associated with antibiotic use.^{17,18}

The rate of antibiotic prescribing is high in Malawi.¹⁹ On the other hand, the high burden of antimicrobial resistance is also a concern. On average, 32% of common bacterial isolates have been found to be resistant to essential antibiotics such as cotrimoxazole, gentamicin and ciprofloxacin.²⁰ This may among other causes be attributed to inappropriate use of antibiotics including non-adherence to treatment.^{21–23} The occurrence of ADRs is one of the factors associated with poor adherence to medicines as they affect the patients' quality of life and loss of trust in the health care professionals.^{24,25} Efforts aimed at identifying and reducing ADRs require a strong Pharmacovigilance (PV) system in a country. PV policies at both the regulatory level and clinical practice need to be primarily informed by local medicine safety data. However, this is not usually the case especially in LMICs such as Malawi as there is a lack of evidence to determine the burden and determinants of ADRs. In this study, we assessed the occurrence and characterized the ADRs associated with antibiotic medicines in the southern Malawi.

Methodology

Study Design and Setting

A retrospective review of medical records was employed in medical wards of public hospitals between June 2022 and October 2022. Three districts (Zomba, Machinga and Nsanje) were randomly selected in southern region of Malawi using the RAND function in excel. Machinga and Nsanje District Hospitals represented the secondary level of care. For Zomba district, there is no secondary level hospital. The largest facility is a tertiary level hospital, the Zomba Central Hospital which acts as a referral health facility for the South-East zone of the country.

Study Population and Sample Size

Our study population were adult patients (\geq 18 years) who were administered one or more antibiotic medicines during their hospital stay. We targeted patients who were hospitalized in the medical wards as most adult patients with infectious diseases are hospitalized in these general medical wards. A total of 304 case management files were included in the study

of which 138 were from Zomba, 84 from Machinga and 82 from Nsanje. The large number from Zomba was as a result of the hospital being a referral health facility for the south-eastern region of Malawi with a bed capacity of 680, unlike Machinga and Nsanje whose bed capacity per facility is 300 patients.^{26,27}

Data Collection

Data were abstracted from case management files of eligible patients using a structured questionnaire. Demographic and clinical data were collected. We applied the global trigger tool for the detection of adverse events (AEs). This tool was developed by the Institute for Healthcare Improvement in the United States of America to help optimize the retrospective detection of adverse events using inpatient hospital records. It applies the use of certain triggers or clues such as switching or ordering of new medicines, abrupt medication stops, abnormal vital signs or laboratory results, and changes in patient prognosis.²⁸ We used the WHO definition of an ADR;

A response to a drug that is noxious and unintended and occurs at doses normally used in man for prophylaxis, diagnosis, or therapy of a disease or for modification of physiological function.

²⁹ Investigators who were experienced pharmacists, laboratory scientists and a medical doctor identified suspected adverse drug reactions (sADRs) by searching for triggers such as the administration of anti-emetic medicines, and any documented events that were experienced by patients after the administration of antibiotic medicines. Further evaluation of the sADRs was also done by the investigators, with one of the investigators, FC, being a senior leader in pharmacovigilance issues in Malawi.

All suspected ADRs (sADRs) were subjected to causality assessment using the Naranjo criteria.³⁰ The Naranjo criteria is an algorithm that uses weighted questions to categorize AEs as definite, probable, possible and unlikely or doubtful ADRs based on the gathered clinical information.³¹ Suspected ADRs were further assessed for seriousness and preventability using the WHO classification³² and the Schumock and Thornton scale³³ respectively.

Statistical Analysis

Data were entered and coded in Microsoft Excel and exported to STATA 14.1 for statistical analysis. The occurrence of sADRs among various patient characteristics was described in terms of frequencies and percentages while continuous variables such as length of hospital stay, and number of medicines administered were summarized in terms of means, medians, and interquartile ranges. The Chi-square test was used to determine the association between various categorical variables such as sex and the occurrence of sADRs. To further assess the strength of association between various independent variables and the occurrence of sADRs, we applied the logistic regression analysis. We used both univariate and multivariate analysis where odds ratios were adjusted by controlling for significant factors using reverse elimination in the regression model.^{34,35} A P value of < 0.05 was considered statistically significant.

Results

Patients' Demographic Characteristics

Of the 304 patients included in the study, 138 (45.4%) were from Zomba Central Hospital, 84 (27.6%) from Machinga District Hospital, and 82 (27%) from Nsanje District Hospital. Among these patients, 48% (n=148) were female, while 52% (n=158) were male (Table 1). The participant's age range was 18–96 years old. Overall, the median age was 44.5 years (IQR 30–62 years) of which for female patients was 43 years (IQR 30–61 years) and for male patients was 45 years (IQR 30–62 years). Fifty-three (17.4%) patients were HIV-positive, 51.6% were HIV-negative and 30.9% were with unknown HIV status. Among the 53 HIV-positive patients, 22.6% had other co-morbidities while 23.5% of the 251 patients who were HIV-negative or with unknown status had other co-morbidities. Common co-morbidities among patients were hypertension (15.1%), type II diabetes mellitus (5.9%) and asthma (2.3%). Diagnosis was based on clinical assessment in 43.2% of the patients while full blood count (FBC) was used to support diagnosis for 56.8% of the patients. No laboratory confirmation by either bacterial culture or anti-microbial sensitivity test was conducted in any of the patients in this study. The common diagnoses were sepsis (24.3%), pneumonia (19.7%), meningitis (4.9%), Cellulitis (3.6%) and peptic ulcers (2.9%).

Category	Sub-group or characteristic	Zomba Central Hospital, n (%) N=138	Machinga District Hospital, n (%) N=84	Nsanje District Hospital, n (%) N=82	Total (N =304)
Age	< 65	102 (42.3)	69 (28.6)	70 (29.1)	241
	≥ 65	36 (57.1)	15 (23.8)	12 (19.1)	63
Sex	Female	79 (54.1)	21 (14.4)	46 (31.5)	146
	Male	59 (37.3)	63 (39.9)	36 (22.8)	158
Co-morbidities (ICD code)	HIV (B24)	33 (62.3)	10 (18.9)	10 (18.9)	53
	Hypertension (110)	38 (82.6)	4 (8.7)	4 (8.7)	46
	Diabetes Mellitus (EI I.8)	12 (66.7)	5 (27.8)	I (5.6)	18
	Asthma (J45.9)	6 (85.7)	(4.3)	0 (0)	7
	Others	10 (66.7)	3 (20)	2 (13.3)	15
Diagnosis (ICD code)	Sepsis (A41.9)	39 (52.7)	12 (16.2)	23 (31.1)	74
	Pneumonia (J18.9)	34 (56.7)	13 (21.7)	13 (21.7)	60
	Meningitis (G00.9)	9 (60.0)	4 (26.7)	2 (13.3)	15
	Cellulitis (L03.9)	3 (27.3)	4 (36.4)	4 (36.4)	П
	Peptic ulcers (K27.9)	5 (55.6)	2 (22.2)	2 (22.2)	9
	Others	48 (35.6)	49 (36.3)	38 (28.2)	135
Tests conducted	FBC	103 (59.9)	29 (16.9)	40 (23.3)	172
	Bacterial culture	0 (0)	0 (0)	0 (0)	0
	Antibiotic sensitivity	0 (0)	0 (0)	0 (0)	0
No. of antibiotics prescribed	I	104 (63.0)	33 (20.0)	28 (17.0)	165
	2	34 (29.8)	38 (33.3)	42 (36.8)	114
	3	0 (0)	13 (56.5)	10 (43.5)	23
	4	0 (0)	0 (0)	2 (100)	2
No. of concomitant	0	2 (33.3)	(6.7)	3 (50.0)	6
medicines	1	29 (32.6)	32 (36.0)	28 (31.5)	89
	2	28 (32.9)	28 (32.9)	29 (24.1)	85
	3	43 (56.6)	17 (22.4)	16 (21.1)	76
	4	36 (75.0)	6 (12.5)	6 (12.5)	48

Abbreviations: ICD, International Classification of diseases 2023 (https://icd.who.int/browse10/2019/en; accessed on 10/05/2023).

Occurrence of Suspected Adverse Drug Reactions

We detected suspected adverse drug reactions (sADRs) in 24% (73/304) of the patients. The prevalence of sADRs for Zomba Central Hospital was 23.9% while for Machinga and Nsanje District Hospitals was 25% and 23.2% respectively (Table 2). The median age for the patients with no sADRs was 42 years (IQR 30–60 years) while for the patients with sADRs, the median age was 48 years (IQR 36–70 years). The patient age group was significantly associated with occurrence of sADRs, p < 0.001. The

Variable	Characteristic	Cases With no sADRs n (%)	Cases With sADRs n (%)	P Value
Age	< 65	193 (80.1)	48 (19.9)	0.001*
	≥ 65	38 (60.3)	25 (39.7)	
Sex	Female	112 (76.7)	34 (23.3)	0.776
	Male	116 (74.8)	39 (24.7)	
HIV status	Negative	189 (76.2)	59 (23.8)	0.652
	Positive	39 (73.6)	14 (26.4)	
Co-morbidities	Non-hypertensive	191 (74.0)	67 (26.0)	0.059
	Hypertensive	40 (87.0)	6 (13.0)	
	Non-diabetic	218 (76.2)	68 (23.8)	0.7
	Diabetic	13 (72.2)	5 (27.8)	
	Non-asthmatic	224 (75.4)	73 (24.6)	0.132
	Asthmatic	7 (100)	0 (0)	
Number of antibiotics prescribed	I	133 (80.6)	32 (19.4)	0.04*
	>	98 (70.5)	41 (29.5)	
Number of concomitant medicines	≤	23 (92.0)	2 (8.0)	0.049*
	>	207 (74.5)	71 (25.5)	
Length of stay (days)	≤ 3	106 (89.8)	12 (10.2)	<0.001*
	>3	125 (67.2)	61 (32.8)	
Hospital	Zomba Central	105 (76.1)	33 (23.9)	0.962
	Machinga District	63 (75.0)	21 (25.0)	
	Nsanje District	63 (76.8)	19 (23.2)	

Table	2	Prevalence	of	Suspected	Adverse	Drug	Reactions	(sADRs)	for	Various	Patient
Charac	teri	stics									

Notes: Chi-square test was used to determine the association between various categorical variables and the occurrence of sADRs. *P value ≤ 0.05 was considered statistically significant. **Abbreviation**: sADRs, suspected adverse drug reactions.

prevalence of sADRs was 23.3% among female and 24.7% among male patients. Among the HIV-negative patients, the prevalence of sADRs was 23.5% while for the HIV-positive patients, the prevalence was 26.4%. Suspected ADR prevalence for hypertensive and diabetic was 13% and 27.8% respectively while none of the asthmatic patients experienced an ADR. For the number of medicines prescribed to patients, the median number of antibiotics was one for patients with no sADRs while for patients with sADRs the median number of antibiotics prescribed was two. For the other concomitant medicines, the median number of medicines was two for both patients with and without sADRs. The median length of hospital stay (LoS) for patients without ADRs was 4 days (IQR 3–6 days) while for patients with ADRs, the median LoS was 6 days (IQR 4–9 days). Occurrence of sADRs was significantly associated with number of antibiotic medicines (p < 0.049) and patient LoS (p < 0.001).

Figure 1 illustrates sADRs according to anatomical system classifications. Most of the sADRs were gastrointestinal events (42.5%), followed by: musculoskeletal (26.3%); cardiovascular (16.3%); central nervous system (13.8%; and urinary events (1.3%).



Figure I Occurrence of suspected adverse drug reactions (s(ADRs) by body systems. The sADRs were classified according to the anatomical system and presented in terms of percentages.

In this study, Ceftriaxone (47.6%) was the most prescribed antibiotic medicine followed by Metronidazole (22.9%), benzylpenicillin (13.2%), gentamicin (6.5%), Amoxicillin \pm clavulanic acid (4.7%), ciprofloxacin (1.7%), flucloxacillin (1.1%), doxycycline (0.4%) and azithromycin (0.2%). By chemical classification, most of these antibiotics were betalactam (66.7%) and nitroimidazole antibiotics (22.9%) while only 10.4% of the antibiotics comprised of other classes. Occurrence of sADRs was proportional to the frequency of prescribing a particular antibiotic, p < 0.001. In total, there were 63 sADRs associated with ceftriaxone, 23 with metronidazole, 19 with benzylpenicillin, 15 with gentamicin and the rest of the sADRs were associated with amoxicillin \pm clavulanic acid, azithromycin, ciprofloxacin, cotrimoxazole and flucloxacillin (Table 3). Common sADRs were abdominal pain (16 events) associated with amoxicillin, benzylpenicillin, ceftriaxone, co-trimoxazole, gentamicin and metronidazole; painful legs (7 events) associated with benzylpenicillin,

Antibiotic Name	ATC Classification	Frequency of Prescriptions (%)	Frequency of sADRs Cases	Adverse Event Detected (Frequency)
Amoxicillin	J01CA04	20 (4.3)	4	Abdominal pain (1), body weakness (1) diarrhea (1), tachycardia (1)
Amoxicillin/ Clavulanic acid	J01CR02	2 (0.4)	I	Hypertension (I)
Azithromycin	J01FA10	I (0.2)	0	
Benzylpenicillin	J01CE01	61 (13.2)	19	Abdominal pain (4), Hematuria (1), body weakness (1), diarrhea (1), joint pain (1), anxiety (1), dizziness (2), joint pain (1), painful legs (1), tachycardia (3), hypotension (1) vomiting (2)
Ceftriaxone injection	J01DD54	220 (47.6)	63	Abdominal pain (12), vomiting (1), irritability (1), back pain (2), body weakness (2), bradycardia (1), chest pain (2), swollen face (1), confusion (1), convulsions (5), dizziness (2), diarrhea (2), dysphagia (1), headache (1), hypertension (3), hypotension (2), joint pain (1), loss of appetite (1), numbness of legs (3), painful legs (6), tachycardia (5), vomiting (7)

 Table 3 Suspected Adverse Drug Reactions and Their Associated Antibiotic Medicines

(Continued)

Antibiotic Name	ATC Classification	Frequency of Prescriptions (%)	Frequency of sADRs Cases	Adverse Event Detected (Frequency)
Ciprofloxacin	J01MA02	8 (1.7)	I	Hypertension (I)
Cotrimoxazole	J01EE01	7 (1.5)	3	Abdominal pain (1), chest pain (1), hypotension (1),
Doxycycline		2 (0.4)	0	
Flucloxacillin	J01CF05	5 (1.1)	3	Anxiety (1), painful legs (2)
Gentamicin	J01GB03	30 (6.5)	15	Abdominal pain (4), hematuria (1), irritability (1), body weakness (1), convulsions (1), dizziness (2), painful legs (1), tachycardia (3), hypotension (1),
Metronidazole	J01XD01	106 (22.9)	26	Abdominal pain (6), anxiety (1), back pain (1), body weakness (1), chest pain (1), convulsions (4), diarrhea (2), dysphagia (1), hypertension (1), painful legs (1), tachycardia (2), vomiting (5)

Table 3 (Continued).

Abbreviations: ATC, Anatomical, Therapeutic and Chemical classification; sADRs, suspected Adverse Drug Reaction.

ceftriaxone, flucloxacillin, gentamicin and metronidazole; vomiting (8 events) associated with benzylpenicillin, ceftriaxone and metronidazole; tachycardia (7 events) associated with amoxicillin, benzylpenicillin, ceftriaxone, gentamicin and metronidazole; and convulsions (5 events) associated with ceftriaxone, gentamicin and metronidazole.

Causality, Seriousness, and Preventability of sADRs

Available clinical data was used to further assess the detected sADRs for causality, seriousness, and preventability. All AEs had a plausible temporal relationship with the culprit antibiotic medicine. Figure 2 provides details for the ADR classification. In terms of causality, 1.4% (n=1) of sADR were definite, 15.1% (n=11) were probable and 83.6% (n =71) were possible ADRs. Serious sADRs were 27.4% (n =20) while non-serious sADRs were 72.6% (n=53). These events were categorized as serious as they were either life threatening or prolonged hospitalization or both. Serious sADRs included convulsions (n = 5), bradycardia (n=1), tachycardia (n=7), hematuria (n=1), swollen face (n=1), hypertension (n=3) and hypotension (n=2). Among the sADRs, 26% (n=19) were preventable while 74% (n=54) were not preventable.



Figure 2 ADR Classification according to causality, seriousness, and preventability. The Naranjo scale, the World Health Organization classification and the Schumock and Thornton scale were applied to perform causality, seriousness and preventability assessments respectively.

Factors Associated with Occurrence of sADRs

A logistic regression analysis was conducted to determine the factors associated with occurrence of sADRs. For multivariate analysis, we adjusted the odd ratios by controlling for significant factors using reverse elimination in the regression model. Occurrence of sADRs was significantly associated with patient age, number of antibiotic medicines taken, and length of hospital stay (Table 4). The geriatric age group (\geq 65 years) was more likely to experience sADRs as compared to the younger age group, with adjusted odds ratio (aOR) of 4.53, 95% CI (2.21–9.28), p < 0.001. By hospital facilities, the results were significant for all the three facilities with aOR of 3.38, 95% CI (1.17–9.76), p< 0.024, aOR 12.97, 95% CI (2.29–73.4), p< 0.004, and aOR 6.95, 95% CI (1.43–33.76), p< 0.016, for Zomba, Machinga and Nsanje respectively. In terms of number of antibiotics administered, patients taking more than one antibiotic medicine had a higher risk of developing sADRs as compared to patients who were on one type of antibiotic medicine, aOR 2.14, 95% CI (1.18–3.90), p < 0.012. These results were not statistically significant when stratified by hospital facility. A hospital stay of > 3 days was also associated with a higher risk of sADRs, with aOR 5.11, 95% CI (2.47–10.55), p < 0.001 as compared to \leq 3 days. By hospital facilities, the results were significant for all facilities with aOR 4.74, 95% CI (1.44–15.61), p< 0.010, aOR 19.09, 95% CI (2.37–15.77), p< 0.006, aOR 3.67, 95% CI (1.08–12.44), p< 0.037, for Zomba, Machinga and Nsanje respectively.

Discussion

We conducted this study in three districts of Malawi to assess the occurrence and characteristics of adverse drug reactions associated with antibiotic medicines. We found sADR prevalence of 24% among the study participants. A study done in Uganda found a 19% burden of antibiotic-associated ADRs among hospitalized patients. This study collected prospective

Variable	Characteristic	aOR	95% CI	P Value
Age	< 65	Ι		
	≥ 65	4.53	2.21–9.28	<0.001
Sex	Female	Ι		
	Male	1.07	0.63-1.82	0.800
HIV status	Non-reactive	Ι		
	Reactive	1.28	0.64–2.54	0.481
Co-morbidities	Non-hypertensive	Ι		
	Hypertensive	0.19	0.07–0.54	0.002
	Non-diabetic	Ι		
	Diabetic	1.29	0.44–3.78	0.647
Number of antibiotics prescribed	I	I		
	>	2.14	1.18–3.9	0.012
Number of concomitant medicines	≤			
	>	2.82	1.39–5.74	0.004
Length of stay (days)	≤ 3			
	>3	5.11	2.47-10.55	<0.001

 Table 4 Factors Associated with Occurrence of Adverse Drug Reactions

Notes: A multivariate logistics regression model was used to assess the strength of association between various independent variables and the occurrence of sADRs. A P value ≤0.05 was considered statistically significant. Abbreviations: aOR, adjusted odds ratio, Clm, confidence interval.

data from both the medical and gynecological wards and was limited to only a tertiary level hospital.¹⁷ In our study, we applied the global trigger tool which is reported to improve the detection rate of adverse events as compared to use of only clinical assessments which are done during routine patient care.³⁶ The global trigger tool has also been used in the detection of 221 ADRs among 1746 (12.7%) patients in a pediatric ward in China³⁷ and 62 ADRs out of 463 patients (13.4%) in the emergency department in India.³⁸

We used the Naranjo criteria for causality assessment and found only one adverse event which was a definite ADR. This was a case of convulsions following administration of ceftriaxone. Patient records indicated that a similar incident had previously occurred in the patient. In addition, there was a positive de-challenge and re-challenge of the reaction, which is not normally done in clinical practice as observed in the rest of the cases. The majority of the sADRs were therefore classified as possible (83.6%) or probable ADRs (15.1%) according to the Naranjo score. The Naranjo criteria provides a simple and reproducible tool for ADR causality assessment.³⁹ However, the major challenge is that it is not possible to respond to all the necessary questions provided by the algorithm where limited patient information is available. This renders most of the potential ADRs rated with a low score.⁴⁰ As noted in our causality assessments, we found very few events which were characterized as definite or possible ADRs since there was limited information to completely rule out other possible causes of the adverse events. Since our patient records were paper based, we encountered challenges such as missing sections in the patient files which not only affects the accurate detection of ADRs but also the subsequent ADR assessments.

Among the detected sADRs, 27.4% were serious. These were central nervous and cardiovascular events such as convulsions, tachycardia, and hypotension. The events met the criteria for seriousness as they were either life threatening or prolonged hospitalization of patient.³² Serious ADRs have worrisome clinical and economic consequences as they require critical patient care to prevent potential loss of life. A study conducted in South Africa found out that the cost of managing ADRs in patients on Tuberculosis (TB) treatment rendered the total cost 17 times higher than the actual cost of TB treatment.⁴¹ In our study, the treatments for sADRs were not quantified as most of these sADRs were not recognized by health care professionals. We also noted a high prevalence of preventable sADRs which contributed to 26% of the cases. Preventability of an ADR is assessed based on the presence of a medication error or any action that would otherwise be avoidable such as administering the wrong dose or lack of monitoring of patients with a known ADR risk.³³ Lack of evidence-based diagnosis may contribute to a higher rate of wrong prescribing of antibiotics in Malawi which may increase the incidence of preventable ADRs. For instance, we observed that no bacterial culture or antimicrobial sensitivity tests were conducted in all patients in this study. FBC was the only test used to support the diagnosis of 56.8% (Table 1) of the patients despite the poor sensitivity and specificity for diagnosis of bacterial infections.^{42–44} Furthermore, the choice of broad-spectrum antibiotics was noted to be very high which may increase the risk of antimicrobial resistance.⁴⁵ Lack of capacity to conduct bacterial culture and antimicrobial sensitivity tests limits the precise selection of the most appropriate narrow- spectrum antibiotic, hence clinicians opt for empirical therapy with broad-spectrum antibiotics.⁴⁶

Occurrence of sADRs was significantly associated with geriatric age (\geq 65 years), number of antibiotic medicines and LoS. These are consistent risk factors for ADR.⁴⁷ Physiological changes such as renal and hepatic impairment are common in geriatric patients.⁴⁸ Usually, these conditions principally affect the elimination of medicines and therefore render geriatric age groups at risk of developing ADRs. Even though the geriatric group had a higher prevalence of sADRs in our study, available patient records did not reveal much information about the presence of age-related co-morbidities such as renal impairment.

Polypharmacy increases not only the drug exposure to patients, but also the potential for drug-drug interactions.⁴⁹ Common antibiotic combinations in this study were ceftriaxone and metronidazole (58 cases); benzylpenicillin and gentamicin (21 cases); benzylpenicillin and metronidazole (9 cases); and metronidazole and amoxicillin (8 cases). There is however limited information about known drug-drug interactions between these antibiotic combinations.⁵⁰ A long LoS is usually associated with a longer duration of treatment, hence increased risk of occurrence of ADRs.⁴⁵ On the other hand, a long LoS is also a consequence of occurrence of an ADR as the affected patients require additional treatment and monitoring.^{51,52} We lacked evidence to make this determination as the days on which sADRs manifested were not recorded in our study.

PV systems seek to identify safety signals.⁵³ These can be previously unknown ADRs or changes in certain aspects of already known ADRs such as presentation or frequency of occurrence that require further investigations.⁵⁴ The majority of the sADRs detected in this study have been previously documented to be associated with antibiotic medicines. To our knowledge, limited information is available for events such as painful legs and joints, tachycardia, and hypotension (Table 3). Convulsions could also be a potential signal as they are observed to occur at a higher frequency in Malawi than expected (<1/1000).^{55,56}

Study Strengths and Limitations

Limitations of this study include measurement bias due to poor documentation or missing information in patient records which might have affected the accuracy of detection and assessment of sADRs. Extremely poor records which were missing crucial data such as medication charts and patient demographics were, however, excluded from the study. Study findings were based on retrospective data. This may underestimate the prevalence of ADRs as we did not directly interview the patients to further explore their personal experiences after taking the medicines. Some of these medicine-related problems may not be documented by clinicians and nurses.⁵⁷ Furthermore, there was a limitation of sample size per health facility. Suspected ADRs in this study may not have been detected or reported to the national PV centre by healthcare providers or patients. This is mostly due to the lack of PV awareness and skills among healthcare workers.⁵⁸ Furthermore, we did not interview possible informants where more information was required for the assessment of sADRs. Based on the available patient data, we precisely assessed the sADRs in terms of causality using the Naranjo criteria. This algorithm has been used in several other studies and reported to be easy to use and reproducible.³⁹

Conclusion

We used the global trigger tool to determine the prevalence of sADRs using retrospective data from patient files. We found a higher prevalence of sADRs associated with antibiotic medicines than reported elsewhere. Furthermore, the number of serious events was high which is a concern regarding the achievement of optimal antibiotic treatment outcomes. This may, among others, contribute to high patient mortality, poor treatment adherence, complicated treatment strategies, antibiotic resistance and increased cost of care.

Abbreviations

ADR, Adverse drug Reactions; sADR, suspected Adverse drug Reactions; ATC, Anatomical Therapeutic Chemical Classification; PV, Pharmacovigilance; WHO, World Health Organization; LMICs, low- and middle-income countries; IQR, Interquartile Range; FBC, Full Blood Count; LoS, Length of Hospital Stay; aOR, adjusted Odds Ratio; CI, Confidence Interval.

Data Sharing Statement

The datasets used and/or analyzed during the current study are available from the corresponding author on reasonable request.

Ethics Approval and Consent to Participate

Approval was granted by the Institution Review Board of Kamuzu University of Health Sciences-Malawi (College of Medicine Research and Ethics Committee (COMREC) under study number P.10/21/3447). In addition, permission was sought from the hospital director or directorates of health and social services in Zomba, Machinga and Nsanje before data collection. All informed consents were obtained from individual patients or their legal guardian after thorough discussion with them before data collection. The study was approved in line with the principles of the declaration of Helsinki.

Consent to Publication

All the hospitals have provided the permission to publish as long as there is no data that might link to specific patients.

Acknowledgment

We would like to thank COMREC for review and ethical clearance of this study and Hospital Director for Zomba Central Hospital, Directors of Health and Social Services for Zomba, Machinga and Nsanje Districts for granting us permission to conduct the study in their facilities. We are also grateful to the pharmacists, nursing, and data officers of the respective hospitals for supporting us during data collection. We also recognize the contribution of Dr Isaac Banda for assisting in reviewing of patient files and hospital laboratory results.

Author Contributions

All authors made a significant contribution to the work reported, whether that is in the conception, study design, execution, acquisition of data, analysis and interpretation, or in all these areas; took part in drafting, revising or critically reviewing the article; gave final approval of the version to be published; have agreed on the journal to which the article has been submitted; and agree to be accountable for all aspects of the work.

Funding

This study was part of the European and Developing Countries Clinical Trials Partnership 2 (EDCTP2) programme supported by the European Union (under grant number TMA2019CDF-2768 COPSMEDS). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Disclosure

The authors declare that they have no conflict of (competing) interests.

References

- 1. Walter SR, Day RO, Gallego B. et al. The impact of serious adverse drug reactions: a population-based study of a decade of hospital admissions in New South Wales, Australia. *Br J Clin Pharmacol.* 2017;83(2):416–426. doi:10.1111/bcp.13124
- Yadesa TM, Kitutu FE, Deyno S, et al. Prevalence, characteristics and predicting risk factors of adverse drug reactions among hospitalized older adults: a systematic review and meta-analysis. SAGE Open Med. 2021;9:205031212110390. doi:10.1177/20503121211039099
- 3. Insani WN, Whittlesea C, Alwafi H, et al. Prevalence of adverse drug reactions in the primary care setting: a systematic review and meta-analysis. *PLoS One*. 2021;16(5):1–24. doi:10.1371/journal.pone.0252161
- 4. Fattinger K, Roos M, Vergères P, et al. Epidemiology of drug exposure and adverse drug reactions in two Swiss departments of internal medicine. *Br J Clin Pharmacol.* 2000;49(2):158–167. doi:10.1046/j.1365-2125.2000.00132.x
- 5. Montastruc JL, Lafaurie M, de Canecaude C, et al. Fatal adverse drug reactions: a worldwide perspective in the World Health Organization pharmacovigilance database. *Br J Clin Pharmacol*. 2021;87(11):4334–4340. doi:10.1111/bcp.14851
- 6. Giardina C, Cutroneo PM, Mocciaro E, et al. Adverse drug reactions in hospitalized patients: results of the FORWARD (Facilitation of Reporting in Hospital Ward) Study. *Frontiers in Pharmacology*. 2018;9:1–12.
- 7. Pouyanne P, Haramburu F, Imbs JL, et al. Admissions to hospital caused by adverse drug reactions: cross sectional incidence study. *Br Med J*. 2000;320(7241):1036. doi:10.1136/bmj.320.7241.1036
- Edwards IR, Aronson JK. Adverse drug reactions: definitions, diagnosis, and management. Lancet. 2000;356(9237):1255–1259. doi:10.1016/ S0140-6736(00)02799-9
- 9. Sultana J, Cutroneo P, Trifirò G. Clinical and economic burden of adverse drug reactions. J Pharmacol Pharmacother. 2013;4(1_suppl):73-77. doi:10.4103/0976-500X.120957
- 10. Osanlou R, Walker L, Hughes DA, et al. Adverse drug reactions, multimorbidity and polypharmacy: a prospective analysis of 1 month of medical admissions. *BMJ Open*. 2022;12(7):1–7. doi:10.1136/bmjopen-2021-055551
- 11. Sulis G, Adam P, Nafade V, et al. Antibiotic prescription practices in primary care in low- And middle-income countries: a systematic review and meta-analysis. *PLoS Med.* 2020;17(6):1–20. doi:10.1371/journal.pmed.1003139
- 12. Siachalinga L, Godman B, Mwita JC, et al. Current antibiotic use among hospitals in the sub-Saharan Africa region; findings and implications. *Infection and Drug Resistance*. 1909;1:0–3.
- Klein EY, Van Boeckel TP, Martinez EM, et al. Global increase and geographic convergence in antibiotic consumption between 2000 and 2015. Proc Natl Acad Sci U S A. 2018;115(15):E3463–E3470. doi:10.1073/pnas.1717295115
- 14. Llor C, Bjerrum L. Antimicrobial resistance: risk associated with antibiotic overuse and initiatives to reduce the problem. *Ther Adv Drug Saf.* 2014;5(6):229–241. doi:10.1177/2042098614554919
- 15. Mohsen S, Dickinson JA, Somayaji R. Update on the adverse effects of antimicrobial therapies in community practice. *Can Fam Physician*. 2020;66(9):651–659.
- 16. Tamma PD, Avdic E, Li DX, et al. Association of adverse events with antibiotic use in hospitalized patients. JAMA Intern Med. 2017;177 (9):1308-1315. doi:10.1001/jamainternmed.2017.1938
- 17. Kiguba R, Karamagi C, Bird SM. Antibiotic-associated suspected adverse drug reactions among hospitalized patients in Uganda: a prospective cohort study. *Pharmacol Res Perspect*. 2017;5(9237):1–16. doi:10.1002/prp2.298

- 18. Jo HG, Jeong K, Ryu JY, et al. Fatal events associated with adverse drug reactions in the Korean national pharmacovigilance database. *J Pers Med.* 2021;12(1):5. doi:10.3390/jpm12010005
- 19. Kayambankadzanja RK, Lihaka M, Barratt-Due A, et al. The use of antibiotics in the intensive care unit of a tertiary hospital in Malawi. *BMC Infect Dis.* 2020;20(1):1–7. doi:10.1186/s12879-020-05505-6
- 20. Kumwenda P, Adukwu EC, Tabe ES, et al. Prevalence, distribution and antimicrobial susceptibility pattern of bacterial isolates from a tertiary Hospital in Malawi. *BMC Infect Dis.* 2021;21(1):1–10. doi:10.1186/s12879-020-05725-w
- 21. Gebeyehu E, Bantie L, Azage M. Inappropriate use of antibiotics and its associated factors among urban and rural communities of Bahir Dar city administration, northwest Ethiopia. *PLoS One*. 2015;10(9):1–14. doi:10.1371/journal.pone.0138179
- 22. Costelloe C, Metcalfe C, Lovering A, et al. Effect of antibiotic prescribing in primary care on antimicrobial resistance in individual patients: systematic review and meta-analysis. *BMJ*. 2010;340(may18 2):1120. doi:10.1136/bmj.c2096
- 23. Endashaw Hareru H, Sisay D, Kassaw C, et al. Antibiotics non-adherence and its associated factors among households in southern Ethiopia. SAGE Open Med. 2022;10:205031212210904. doi:10.1177/20503121221090472
- 24. Rolfes L, van Hunsel F, Taxis K, et al. The impact of experiencing adverse drug reactions on the patient's quality of life: a retrospective crosssectional study in the Netherlands. Drug Saf. 2016;39(8):769-776. doi:10.1007/s40264-016-0422-0
- 25. Berhe DF, Taxis K, Haaijer-Ruskamp FM, et al. Impact of adverse drug events and treatment satisfaction on patient adherence with antihypertensive medication a study in ambulatory patients. *Br J Clin Pharmacol.* 2017;83(9):2107–2117. doi:10.1111/bcp.13312
- 26. Makwero MT. Delivery of primary health care in Malawi. African J Prim Heal Care Fam Med. 2018;10:1-3.
- 27. Kröner-Fresenius-Stiftung E. Capacity building in physiotherapy for the south-eastern region of Malawi in Zomba as its centre, https://www.ekfs. de/en/humanitarian-funding/currently-funded-projects/capacity-building-in-Physiotherapy. (2021, accessed 24, October 2023.).
- 28. Griffin F, Resar R. IHI global trigger tool for measuring adverse events. *IHI Innov Ser white Pap.* 2009;1:1–44.
- 29. Montané E, Santesmases J. Adverse drug reactions. Med Clini. 2020;154(5):178-184. doi:10.1016/j.medcli.2019.08.007
- 30. Naranjo CA, Busto U, Sellers EM, et al. A method for estimating the probability of adverse drug reactions. *Clin Pharmacol Ther.* 1981;30 (2):239–245. doi:10.1038/clpt.1981.154
- 31. Bethesda MD. Adverse drug reaction probability scale (naranjo) in drug induced liver injury. *Natl Institutes Heal US Dept Heal Hum Serv.* 2019;1:1–5.
- 32. World_Health_Organisation. Safety of Medicines: a guide to detecting and reporting adverse drug reactions. *World Health Organization Geneva*, (2002); Available from: https://apps.who.int/iris/handle/10665/67378. Accessed July 17, 2024.
- 33. Schumock GT, Thornton JP. Focusing on the preventability of adverse drug reactions. Hosp Pharm. 1992;27(6):538.
- 34. Shegena EA, Nigussie KA, Tamukong R, et al. Prevalence and factors associated with adverse drug reactions among heart failure patients hospitalized at Mbarara Regional Referral Hospital, Uganda. *BMC Cardiovasc Disord*. 2022;22(1):1–13. doi:10.1186/s12872-022-02937-7
- 35. Alomar MJ. Factors affecting the development of adverse drug reactions (Review article). Saudi Pharm J. 2014;22(2):83-94. doi:10.1016/j. jsps.2013.02.003
- 36. Noorda NMF, Sallevelt BTGM, Langendijk WL, et al. Performance of a trigger tool for detecting adverse drug reactions in patients with polypharmacy acutely admitted to the geriatric ward. *Eur Geriatr Med.* 2022;13(4):837–847. doi:10.1007/s41999-022-00649-x
- 37. Ji HH, Song L, Xiao JW, et al. Adverse drug events in Chinese pediatric inpatients and associated risk factors: a retrospective review using the Global Trigger Tool. *Sci Rep.* 2018;8(1):1–7. doi:10.1038/s41598-018-20868-2
- Pandya AD, Patel K, Rana D, et al. Global trigger tool: proficient adverse drug reaction autodetection method in critical care patient units. *Indian J Crit Care Med.* 2020;24(3):172–178. doi:10.5005/jp-journals-10071-23367
- 39. Khan LM, Al-harthi SE, Osman AM, et al. Dilemmas of the causality assessment tools in the diagnosis of adverse drug reactions. *Saudi Pharm J*. 2016;24(4):485–493. doi:10.1016/j.jsps.2015.01.010
- 40. Seger D, Barker K, McNaughton C. Misuse of the naranjo adverse drug reaction probability scale in toxicology, *Clin Toxicol*. 2013;6. doi:10.1016/j. earlhumdev.2006.05.022
- 41. Knight LK, Lehloenya RJ, Sinanovic E, et al. Cost of managing severe cutaneous adverse drug reactions to first-line tuberculosis therapy in South Africa. *Trop Med Int Heal.* 2019;24(8):994–1002. doi:10.1111/tmi.13275
- Martínez-Domínguez J, Sierra-Martínez O, Galindo-Fraga A, et al. Antibiotic prescription errors: the relationship with clinical competence in junior medical residents. BMC Med Educ. 2022;22(1):1–14. doi:10.1186/s12909-022-03499-0
- 43. Cruz AT, Mahajan P, Bonsu BK, et al. Accuracy of complete blood cell counts to identify febrile infants 60 days or younger with invasive bacterial infections. JAMA Pediatr. 2017;171(11):1-8. doi:10.1001/jamapediatrics.2017.2927
- 44. Günay Y, Taşdöven İ, Kozan R, et al. Investigation of predictive value of complete blood count in the diagnosis of acute complicated appendicitis. *Haseki Tip Bul.* 2019;57(1):26–31. doi:10.4274/haseki.galenos.2018.4567
- 45. Cižman M, Plankar Srovin T. Antibiotic consumption and resistance of gram-negative pathogens (collateral damage). *GMS Infect Dis.* 2018;6: Doc05. doi:10.3205/id000040
- 46. Melander RJ, Zurawski DV, Melander C. Narrow-spectrum antibacterial agents. Medchemcomm. 2018;9(1):12-21. doi:10.1039/C7MD00528H
- 47. Alhawassi TM, Krass I, Bajorek B, et al. A systematic review of the prevalence and risk factors for adverse drug reactions in the elderly in the acute care setting. *Clin Interv Aging*. 2014;9:2079–2086. doi:10.2147/CIA.S71178
- 48. Muirhead GJ, Wilner K, Colburn W, et al. The effects of age and renal and hepatic impairment on the pharmacokinetics of sildenafil citrate. Br J Clin Pharmacol Suppl. 2002;53(Suppl 1):21S-30S. doi:10.1046/j.0306-5251.2001.00029.x
- 49. Liao PJ, Mao CT, Chen TL, et al. Factors associated with adverse drug reaction occurrence and prognosis, and their economic impacts in older inpatients in Taiwan: a nested case-control study. *BMJ Open.* 2019;9(5):1–10. doi:10.1136/bmjopen-2018-026771
- 50. Drug-interaction-checker. Medscape, https://reference.medscape.com/drug-interactionchecker. (accessed 15, January 2023.).
- 51. Hoogervorst-Schilp J, Langelaan M, Spreeuwenberg P, et al. Excess length of stay and economic consequences of adverse events in Dutch hospital patients. *BMC Health Serv Res.* 2015;15(1):1–7. doi:10.1186/s12913-015-1205-5
- 52. Classen DC. Adverse drug events in hospitalized patients excess length of stay, extra costs, and attributable mortality. JAMA J Am Med Assoc. 1997;277:301.
- 53. Lerch M, Nowicki P, Manlik K, et al. Statistical signal detection as a routine pharmacovigilance practice: effects of periodicity and resignalling criteria on quality and workload. *Drug Saf.* 2015;38(12):1219–1231. doi:10.1007/s40264-015-0345-1

- 54. Insani WN, Pacurariu AC, Mantel-Teeuwisse AK, et al. Characteristics of drugs safety signals that predict safety related product information update. *Pharmacoepidemiol Drug Saf.* 2018;27(7):789–796. doi:10.1002/pds.4446
- 55. Electronic-medicines-compendium. Summary of product characteristics, https://www.medicines.org.uk/emc/. (2023, accessed 15, January 2023.).
- 56. Joint_Formularly_Commitee. British National Formulary 83. London. 2022. doi: 10.1002/jps.3030470133
- 57. Hoffman RS. Understanding the limitations of retrospective analyses of poison center data. Clin Toxicol. 2007;45(8):943-945. doi:10.1080/15563650701233370
- Jusot V, Chimimba F, Dzabala N, et al. Enhancing pharmacovigilance in sub Saharan Africa through training and mentoring: a GSK pilot initiative in Malawi. Drug Saf. 2020;43(6):583–593. doi:10.1007/s40264-020-00925-4

Drug, Healthcare and Patient Safety

Dovepress

Publish your work in this journal

Drug, Healthcare and Patient Safety is an international, peer-reviewed open-access journal exploring patient safety issues in the healthcare continuum from diagnostic and screening interventions through to treatment, drug therapy and surgery. The journal is characterized by the rapid reporting of reviews, original research, clinical, epidemiological and post-marketing surveillance studies, risk management, health literacy and educational programs across all areas of healthcare delivery. The manuscript management system is completely online and includes a very quick and fair peer-review system. Visit http://www.dovepress.com/testimonials.php to read real quotes from published authors.

Submit your manuscript here: https://www.dovepress.com/drug-healthcare-and-patient-safety-journal

f 🄰 in 🕨 DovePress